Author’s response to reviews

Title: Anxiety at Age 15 Predicts Psychiatric Diagnoses and Suicidal Ideation in Late Adolescence and Young Adulthood: Results From Two Longitudinal Studies

Authors:

Sabrina Doering (sabrina.doering@gu.se)
Paul Lichtenstein (Paul.Lichtenstein@ki.se)
Christopher Gillberg (christopher.gillberg@gnc.gu.se)
Christel Middeldorp (c.middeldorp@uq.edu.au)
Meike Bartels (m.bartels@vu.nl)
Ralf Kuja-Halkola (Ralf.Kuja-Halkola@ki.se)
Sebastian Lundström (sebastian.lundstrom@gnc.gu.se)

Version: 2 Date: 30 Oct 2019

Author’s response to reviews:

Response letter can be found under uploaded files &gt; supplementary material ('Response letter_Doering_R2.docx')